Author’s response to reviews

Title: Goal conflict, goal facilitation, and health professionals' provision of physical activity advice in primary care: An exploratory prospective study

Authors:

Justin Presseau (justin.presseau@ncl.ac.uk)
Jill J Francis (j.francis@abdn.ac.uk)
Neil C Campbell (n.campbell@abdn.ac.uk)
Falko F Sniehotta (falko.sniehotta@ncl.ac.uk)

Version: 4 Date: 10 July 2011

Author's response to reviews: see over
Dear Implementation Science Editorial Team,

Upon careful proofreading of our manuscript, we noticed three minor content-related errors.

1) On page 5, in the sentence: "A systematic review of social cognition models applied to predict health professional behaviour identified 16 prospective studies testing the TPB with 1,882 health professionals [10]."], the number 16 should be changed to 14 (i.e., 14 prospective studies).

2) Similarly, on page 16, in the sentence: “Of the 16 prospective studies testing the TPB in health professionals identified by Godin et al.’s [10] systematic review, seven had sample sizes of 50 participants or less.”], the number 16 should be changed to 14 (i.e., 14 prospective studies).

3) On pages 17-18, in the sentence: “In addition, the amount of variance in intention and behaviour accounted for (46% and 48%) was in line with mean frequency-weighted $R^2$ observed across TPB studies reported by Godin et al.’s review [10] (59% and 31%, respectively),"], the number 31% should be changed to 35%.

The updated version of the manuscript reflects these minor changes. Thank you for your time and help with these corrections.

Sincerely,

Justin Presseau